Author's response to reviews

Title: hTERT promoter activity and CpG methylation in HPV-induced carcinogenesis

Authors:

Jillian de Wilde (jilliandw@hotmail.com)
Jan M. Kooter (jan.kooter@falw.vu.nl)
Renée M. Overmeer (r.overmeer@vumc.nl)
Debbie Claassen-Kramer (D.Claassen@vumc.nl)
Chris J.L.M. Meijer (cjlm.meijer@vumc.nl)
Peter J.F. Snijders (pif.snijders@vumc.nl)
Renske D.M. Steenbergen (r.steenbergen@vumc.nl)

Version: 5 Date: 14 May 2010

Author's response to reviews:

Dear Editor,

We have changed the sentence on approval of the study by the Medical Ethical Committee on Page 9, lines 8 to 11, as is highlighted in yellow.

I hope this amendment is in agreement with the BMC Cancer policy.

Sincerely,

Renske Steenbergen